Author's response to reviews

Title: Improved cardiovascular diagnostic accuracy by pocket size imaging device in non cardiologic outpatients. The NaUSiCa (Naples Ultrasound Stethoscope in Cardiology) Study

Authors:

Maurizio Galderisi (mgalderi@unina.it)
Alesandro Santoro (cantopo@hotmail.com)
Marco Versiero (mailpermarco@tiscali.it)
Vincemzo Schiano Lomoriello (capatost@libero.it)
Roberta Esposito (robyeire@tin.it)
Rosa Raia (rosa.78@libero.it)
Francesca Farina (francesca.farina81@gmail.com)
Pier Luigi Schiattarella (pluigi@libero.it)
Manuela Bonito (manu.bonito@otmail.it)
Marinella Olibet (marinellaolibet@libero.it)
Giovanni de Simone (gianni49@libero.it)

Version: 2 Date: 19 November 2010

Author's response to reviews: see over
Naples, November 18, 2010

Dear Dr. Picano,

Enclosed you will find the revised draft of the original manuscript entitled “Improved cardiovascular diagnostic accuracy by pocket size imaging device in non cardiologic outpatients: the NaUSiCa Study” for tentative publication on Cardiovascular Ultrasound.

The manuscript has been revised according to the concerns of both the reviewers.

We hope that our manuscript could be meet the criteria for publication on Cardiovascular Ultrasound and I hope you will agree with us.

Thanking you for attention, I remain.

Sincerely yours.

Maurizio Galderisi, MD, FESC